Author's response to reviews

Title: Reactive Lymphoid Hyperplasia of the Liver in a Patient with Colon Cancer: Report of Two Cases

Authors:

Hiroki Takahashi (takahashi@med.nagoya-cu.ac.jp)
Yoichi Matsuo (matsuo@med.nagoya-cu.ac.jp)
Hirozumi Sawai (sawai@med.nagoya-cu.ac.jp)
Hitoshi Funahashi (funa84@med.nagoya-cu.ac.jp)
Mikinori Satoh (miiki@med.nagoya-cu.ac.jp)
Yuji Okada (yuji@med.nagoya-cu.ac.jp)
Hiroshi Inagaki (hinagaki@med.nagoya-cu.ac.jp)
Hiromitsu Takeyama (takeyama@med.nagoya-cu.ac.jp)
Tadao Manabe (mnb@med.nagoya-cu.ac.jp)

Version: 2 Date: 18 August 2006

Author's response to reviews: see over
Dear: Annabel Phillips PhD, Assistant Editor

Thank you for your kind letter of August 17. I am pleased that you are interested in our paper, and I hope that I have answered the reviewer’s question adequately. I understand that final acceptance depends on satisfactory resolution of these issues.

We appreciate the thoroughness with which the reviewers regarded our paper and hope that it is now suitable for publication in BMC Gastroenterology. I look forward to hearing from you again.

With best regards,
Yours sincerely,

Hiroki Takahashi, MD

Department of Gastroenterological Surgery, Nagoya City University Graduate School of Medical Sciences
Kawasumi 1, Mizuho-cho, Mizuho-ku, Nagoya, Aichi pref. 4678601, Japan
phone: +81-52-853-8226 / fax: +81-52-842-3906
E-mail address
Hiroki Takahashi: takahasi@med.nagoya-cu.ac.jp
Yoichi Matsuo: matsuo@med.nagoya-cu.ac.jp
Hirozumi Sawai: sawai@med.nagoya-cu.ac.jp
Hitoshi Funahashi: funa84@med.nagoya-cu.ac.jp
Mikinori Satoh: miki@med.nagoya-cu.ac.jp
Yuji Okada: yuji@med.nagoya-cu.ac.jp
Hiroshi Inagaki: hinagaki@med.nagoya-cu.ac.jp
Hiromitsu Takeyama: takeyama@med.nagoya-cu.ac.jp
Tadao Manabe: mnb@med.nagoya-cu.ac.jp
Answers to reviewer’s question

#1: As the reviewer suggested, we changed “type-2 tumor” to “An ulcerated tumor with sharply demarcation”.  (Page 5, Lane 3)

#2: We added Abstract in our manuscript.